Author's response to reviews

Title: Assessing Protein Energy Wasting in a Malaysian Haemodialysis Population Using Self-Reported Appetite Rating: A Cross-Sectional Study

Authors:

Sharmela Sahathevan (sham_0901@yahoo.com)
Chee-Hee Se (jeremysch3232@gmail.com)
See-Hoe Ng (seehoe88@gmail.com)
Karuthan Chinna (karuthan@gmail.com)
Gilcharan S Harvinder (harvin3386@yahoo.com)
Winnie SS Chee (winnie_chee@imu.edu.my)
Bak L Goh (bak.leong@gmail.com)
Halim A Gafor (halimgafor@gmail.com)
Sunita Bavanandan (savanandan@gmail.com)
Ghazali Ahmad (ghazaliahmad@moh.gov.my)
Tilakavati Karupaiah (tilly_karu@yahoo.co.uk)

Version: 5 Date: 17 April 2015

Author's response to reviews: see over
17th April 2015

Dr Liffert Vogt  
Associate Editor  
BMC Neprology

Dear Dr. Vogt,

Re: Submission of Revised Manuscript MS ID: 5211732951582230

On behalf of the authors, I wish to thank the Editor for the suggestions to improve the manuscript. We have completed the changes requested by the editors.

We have highlighted in yellow the changes made in the manuscript. We are now submitting the revised manuscript. Thank you for your editorial services and acceptance of our manuscript for publication.

Yours sincerely,

TILAKAVATI KARUPAIAH, PhD, APD, AN  
Associate Professor  
Dietetics Program  
School of Healthcare Sciences  
Faculty of Health Sciences  
National University of Malaysia  
Tel  : +603-92897245 (Office) +6019-2731400 (Mobile)